Author's response to reviews

Title: Seroepidemiology of infection with Toxoplasma gondii in healthy blood donors of Durango, Mexico

Authors:

Cosme Alvarado-Esquivel (alvaradocosme@yahoo.com)
Miguel-Francisco Mercado-Suarez (miguelfranms@hotmail.com)
Alfredo Rodriguez-Briones (cetsdurango@yahoo.com.mx)
Laura Fallad-Torres (alvaradocosme@yahoo.com)
Julio-Octavio Ayala-Ayala (miguelfranms@hotmail.com)
Luis-Jorge Nevarez-Piedra (cetsdurango@yahoo.com.mx)
Ehecatl Duran-Morales (cetsdurango@yahoo.com.mx)
Sergio Estrada-Martinez (sem@mexico.com)
Oliver Liesenfeld (oliver.liesenfeld@charite.de)
Jose-Angel Marquez-Conde (alvaradocosme@yahoo.com)
Sergio-Arturo Martinez-Garcia (sergiomicro@gmail.com)

Version: 4 Date: 2 July 2007

Author's response to reviews: see over

Ref. MS 1189417154132410: “Seroepidemiology of infection with *Toxoplasma gondii* in healthy blood donors of Durango, Mexico”.

Dear Editor,

Attached please find a revised version of our manuscript that had been modified according to the BioMed Central Editorial Production Team’s comment. As recommended, we have removed the Key words section.

Thank you in anticipation for your kind attention.

Kind regards,

Dr. Cosme Alvarado-Esquivel.
Facultad de Medicina
Avenida Universidad y Fanny Anitua
34000 Durango, Dgo. México.
Tel/Fax.: 0052 618 8 128009
Email: alvaradocosme@yahoo.com